Author's response to reviews

Title: Dietary analysis and patterns of nutritional supplement use in normal and age-related macular disease affected subjects: a prospective cross-sectional study

Authors:

Hannah Bartlett (bartlehe@aston.ac.uk)
Frank Eperjesi (F.Eperjesi@aston.ac.uk)

Version: 3 Date: 1 September 2004

Author's response to reviews:

Thank you for your comments. The changes are listed below:

1. The full stop has been removed a the end of the title.
2. The authors’ qualifications have been removed from the title page.
3. The authors have been listed without the email addresses.
4. The references in the text have been un-superscripted.
5. The figure and table placement text has been removed.
6. The table titles have been moved to above the tables.
7. Figure 1 has been cropped in Word.

Kind regards

Hannah Bartlett